Title: Increasing the frequency of hand washing by healthcare workers does not lead to commensurate reductions in staphylococcal infection in a hospital ward.

Authors:

Clive B Beggs (C.B.Beggs@bradford.ac.uk)
Simon J Shepherd (S.J.Shepherd@bradford.ac.uk)
Kevin G Kerr (kevin.kerr@hdft.nhs.uk)

Version: 4 Date: 26 August 2008

Author's response to reviews:

Title - please ensure that the title is identical in the manuscript and the submission system: Done

Abstract - Please ensure that the Abstract is present in manuscript and matches exactly what is in the submission system: Done

Introduction - Please rename this section `Background: Done

Table title and legend - The table title should appear above the table and should be only one sentence long. Please place any other text as a legend (or key), which should appear below the table: Done